Author’s response to reviews

Title: Primary hepatic leiomyoma in a Chinese female patient without underlying disease: a case report

Authors:
Baoxing Jia (andylaputa@qq.com)
Zhe Jin (654295896@qq.com)
Pin Gao (gaopin0718@hotmail.com)
Yahui Liu (liuyahuisci@163.com)

Version: 2 Date: 20 Aug 2019

Author’s response to reviews:

Editor Comments: The language in your manuscript has improved a lot, however, it would still benefit from a final copy-edit from a professional editor or a native English speaking colleague.

R: Thank you for your suggestion. The manuscript has been polished by Medjaden Bioscience Limited, including language certificate.

Reviewer reports:

Paolo Magistri (Reviewer 2): As compared to my previous revision, authors improved significantly the quality of the paper that is now acceptable for publication, besides the intrinsic limits of a case report in terms of scientific relevance.

The table helps a lot the reading and the comprehension of the paper, and images are now more relevant.

R: Thank you very much for your recognition of the quality of the manuscript.

Authors should provide the details of the companies whom products are cited in the text (name of the company, state).

R: Thank you for your suggestion. I have added the details of the companies into manuscript. (Treatment and outcome section, line 5, 12-15, 17, page 5)